Author's response to reviews

Title: Calcium deficiency assessment and biomarkers identification by an integrated urinary metabonomics analysis

Authors:

Maoqing Wang (wang_maoqing@yahoo.com)
Xue Yang (yaya_orange2005@yahoo.com.cn)
Fan Wang (Yifan.701@163.com)
Ran Li (yo_yo1983@126.com)
Hua Ning (ninghua188@126.com)
Lixin Na (nalixin2003@yahoo.com.cn)
Yifan Huang (yyzhyf@hotmail.com)
Yue Song (wolingdong@126.com)
Liyan Liu (yanziliu2100@163.com)
Hongzhi Pan (panhongzhi@163.com)
Qiuju Zhang (zqj81129@sina.com)
Lijun Fan (fanlijun1229@sina.com)
Ying Li (liying_helen@163.com)
Changhao Sun (Sun2002changhao@yahoo.com)

Version: 4 Date: 18 February 2013

Author's response to reviews:

Dear Ursula D'Souza, PhD

Senior Editor,

Thank you very much for the publication of our manuscript. As you suggested, we downloaded the most recent version of our manuscript from the following link: (http://www.biomedcentral.com/media/man/1704763993787749/msd/main/0). In order to make our manuscript conform to your journal style, we revised it point-to-point as detailed below:

1. We ensure that each author affiliation contains, in the following order, department, institution, street name and number, town, post code and country in the revised manuscript.

2. We have deleted the URLs from the main text. We ensure that all references and citations are renumbered accordingly in the revised manuscript.

3. We have provided an overall figure title for Figure 2. The legend for the specific parts is followed after this sentence.

4. We have provided an overall title for additional file 6.

5. We have checked our manuscript for the typographical errors and we ensure that this is the final version of your manuscript.
Please contact us if you have any problems or questions.

Sincerely yours,

Wang Maoqing